Author's response to reviews

Title: Estimation of Hospital Emergency Room Data Using OTC Pharmaceutical Sales and Least Mean Square Filters

Authors:

   Dr Amir-Homayoon Najmi (najmi@jhuapl.edu)
   Steve Magruder (steve.magruder@jhuapl.edu)

Version: 3 Date: 12 Mar 2004

PDF covering letter
I have made all the discretionary changes suggested by reviewer Dr. Buckeridge (see below), and would like to submit the final version of the following paper for on-line publication:

**Estimation of hospital emergency room data using OTC pharmaceutical sales and least mean square filters**

A. H. Najmi and S. F. Magruder

National Security Technology Department
The Johns Hopkins University
Applied Physics Laboratory
Laurel, MD 20723-6099

Contact information:

Dr. A. H. Najmi
Tel: 443-778 3320
Email: najmi@jhuapl.edu

---

I have attempted to make all the formatting changes to the best of my abilities. The only one I have not attempted was the “Authors’ contribution” section; I have no idea what to put in it! Also, please check the “References” section for proper referencing of books and reports.

I have made the following changes to address the discretionary changes suggested by the reviewer (Dr. Buckeridge):

a. In the abstract, there is no reference to being “... not aware of...”, and I have added reference 1 to the effect that “... we are in fact aware of ...”.
b. In the Methods section, I have taken out any negative references to the “smoothing problem”.
c. In the Conclusions section, I have narrowed it to the case under investigation in this paper and avoided any more general assertion.

A. H. Najmi, M.A. (Cantab.), D.Phil.(Oxon.)